Author’s response to reviews

Title: Design of a randomized controlled ACL injury prevention study.

Authors:

Anne Benjaminse (a.benjaminse@med.umcg.nl)
Koen A.P.M. Lemmink (k.a.p.m.lemmink@rug.nl)
Ron L. Diercks (r.l.diercks@orth.umcg.nl)
Bert Otten (e.otten@med.umcg.nl)

Version: 2 Date: 30 March 2010

Author’s response to reviews:

Dear Editor,
Please find attached a revised version of our manuscript with #: 1590778641365298.

Study protocol
Design of a randomized controlled ACL injury prevention study.
Anne Benjaminse, Koen A.P.M. Lemmink, Ron L. Diercks and Bert Otten

The only change we have made is that we have added our trial registration number.

Yours sincerely,
Anne Benjaminse